Author's response to reviews

Title: Obesity and pronated foot type may increase the risk of chronic plantar heel pain: a matched case-control study

Authors:

Damien B Irving (damien_irving@yahoo.com.au)
Jill L Cook (j.cook@latrobe.edu.au)
Mark A Young (markalex@mac.com)
Hylton B Menz (h.menz@latrobe.edu.au)

Version: 5  Date: 16 May 2007

Author’s response to reviews:

May 16, 2007

Reviewer comment:
Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Author response:
Figures have been cropped as closely as possible.